Author's response to reviews

Title: Effectiveness of second generation antipsychotics: A systematic review of randomized trials.

Authors:

Erik Johnsen (erij@ihelse.net)
Hugo A Jorgensen (hugo.jorgensen@psyk.uib.no)

Version: 5 Date: 24 April 2008

Author's response to reviews:

The abbreviations section belonging to Table 5 was displaced to below Conclusion. This was revised so that it was cut and pasted back in its place below Table 5.